



Article The Role of Value in Extending the Lifetime of Products: An Analysis of Perceived Value and Green Consumption Values on Pro-Circular Behaviors of Repair and Reuse

Claudia Arias *🝺, Javier Bernardo Cadena Lozano 🕩 and Miguel Angel Bello Bernal 🕩

School of Business, CESA, Bogotá 110311, Colombia; javier.cadena@cesa.edu.co (J.B.C.L.); miguel.bello@cesa.edu.co (M.A.B.B.)

* Correspondence: claudia.arias@cesa.edu.co; Tel.: +57-1-3395300

Abstract: Promoting circular business models is a clear path to tackling current environmental and social issues. The success of these models depends not only on companies in charge of creating new products, processes, and models that include circular strategies but also on consumers and their choices. Thus, fully understanding consumers and the factors that influence accepting and adopting practices towards circularity is imperative. One of these factors is the relationship that consumers have with the products they purchase and their assessment of these products through the concept of perceived value. This research aimed to explore the relationship between the dimensions of perceived value (i.e., functional, emotional, and social) and pro-circular behaviors of repair and reuse, as well as the role of green consumption values in this relationship. A representative sample of 417 people was surveyed in Medellín (Colombia). Using an Ordinal Logit model, we tested the hypotheses proposed. Findings reveal that functional value is a primary starting point to encourage behaviors aiming to extend the lifetime of products. Moreover, the interaction between sources of value (perceived value and green consumption values) increases the probability of reusing and repairing. Based on these and other relevant results, managerial implications and opportunities for future research are proposed.

Keywords: perceived value; green consumption values; pro-circular behaviors; repair; reuse; circular economy

1. Introduction

Environmental issues demanded new perspectives from several actors to tackle them. One of these perspectives is to change from a linear economic model, "take-make-usedispose", to a circular economy. This change requires all market participants' awareness, knowledge, and engagement [1]. Particularly, businesses and consumers are the two main actors involved in the causes and the solutions of environmental problems because of their production decisions and consumption choices. Thus, both companies and consumers may act together through circular business models. These models are defined as those that use innovation to create, deliver, and capture value to improve resource efficiency and continually reuse products and materials by extending the lifespan of products and by reducing waste, where possible, thereby achieving environmental, social, and economic benefits [2,3].

The success of these models depends not only on companies in charge of creating new products, services, processes, and models that include circular strategies but also on consumers and their choices (e.g., the number of products they buy, the openness to accept products with circular attributes and new business models, how they deal with used products, and adopt pro-circular behaviors) [1]. Thus, the consumer is a key player and an essential piece in achieving the goals of circular economy models [4]. Considering the perspectives of consumers, as those who acquire and use products, as well as understanding all the factors involved in their consumption decisions, is crucial to promoting



Citation: Arias, C.; Cadena Lozano, J.B.; Bello Bernal, M.A. The Role of Value in Extending the Lifetime of Products: An Analysis of Perceived Value and Green Consumption Values on Pro-Circular Behaviors of Repair and Reuse. *Sustainability* **2024**, *16*, 1567. https://doi.org/10.3390/ su16041567

Academic Editor: Colin Michael Hall

Received: 27 December 2023 Revised: 25 January 2024 Accepted: 2 February 2024 Published: 13 February 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). and implementing these circular business models [5–7]. One of these factors is consumers' relationship with the products they purchase based on the concept of perceived value.

The concept of consumer's perceived value, first proposed in marketing research, was defined as "consumer's overall assessment of the utility of a product (or service) based on perception of what is received and what is given" [8] (p.4). Value is the expression of fundamental desires and goals that consumers want to gain [9]. Hence, this concept is considered an essential constituent of relationship marketing and the primary basis for consumer behavior associated with purchasing [10]. Thus, the role of perceived value was oriented to understanding the acquisition phase in the consumption stages. Nevertheless, the new demands and needs for sustainable products and sustainable business models (e.g., circular business models) drive a more holistic perspective from marketing in which other behaviors beyond purchase ones must be analyzed to be promoted (e.g., pro-circular behaviors). The role of marketing in circular business models could be multifaceted and is crucial for the success and adoption of circular economy practices. To accomplish this, gaining a more comprehensive understanding of consumers and the variables that influence their relationship with this kind of behavior is imperative. This research aims to contribute to this goal by exploring the role of perceived value in adopting pro-circular behaviors. To the best of our knowledge, little research addressed this relationship.

From the initial but valuable literature on consumer and circular behaviors, previous research emphasized the relevance of performing analyses oriented to groups of similar behaviors or specific actions for specific products [5]. Among the different circular business models, those focused on practices that aim to extend the useful life of products were pointed out as crucial within a circular economy [11]. The previous literature mainly suggests reusing and repairing behaviors as desirable options because of their potential to reduce the environmental impact across the whole value chain [6,12]. In line with this perspective, the first objective of this research is to examine the role of perceived value in adopting reusing and repairing behaviors. It does this through the lens of two product categories: furniture and clothing. These categories are identified by their environmental impact and importance in everyday life [13].

As already mentioned, perceived value is the primary basis for consumer behavior associated with purchasing [10]. Nevertheless, the prior literature pointed out that there could be other sources of value in a consumption experience [14]. Because we explored behaviors that, by definition, avoid the acquisition and are essential in other stages of consumption (i.e., repairing and reusing), we involved green consumption values as another source of value, stating the second objective of our study: to analyze their role in the relationship between perceived value and pro-circular behaviors. By accomplishing this, we contribute to the current literature on marketing and sustainability by addressing a comprehensive analysis that establishes the influence of products' perception in adopting practices aiming at their circularity and the interaction between products' evaluation and environmental considerations in pro-circular behavior. This approach could be valuable in defining more effective strategies and initiatives, from marketing to promoting circular business models.

The remainder of this paper is structured as follows. Section 2 presents the theoretical framework and definition of the hypotheses. Section 3 shows the methodology used to collect and analyze data. Section 4 presents the main findings of the study, involving descriptive results and hypotheses testing. Section 5 discusses the theoretical implications of the study, including opportunities for future research. Finally, we provide concluding remarks with managerial implications and acknowledge the study's limitations.

2. Theoretical Framework

2.1. Perceived Value, Sustainability, and Circularity

Perceived value is a subjective assessment that a consumer makes about a product or service's worth or benefit [15]. This assessment derives from the trade-off between benefits and sacrifices where the customers interact with a particular good [16]. The benefits involve

extrinsic and intrinsic attributes of the product, and sacrifices are related to perceived costs, including monetary and non-monetary aspects [17–22].

Perceived value is conceptualized as a multidimensional construct [22–24]. Initially, Sheth et al. [23] proposed the Theory of Consumption Value including five dimensions (functional, emotional, social, epistemic, and conditional): functional value relates to the tangible benefits and utility a product or service provides; emotional value is associated with the feelings, emotions, and psychological benefits that a product or service can evoke, including actions against the product; social value refers to benefits derived from an association with reference groups (e.g., a person's social image, standing, relationships, or identity); epistemic value is about the novelty aspect of a product involving the knowledge and information the product or service imparts [21]; conditional value is tied to situational factors and consider how the context, timing, and circumstances affect the perceived worth of a product [21]. After that, Sweeney and Soutar [22] developed a perceived value measurement scale (PERVAL) based on functional value, emotional value, and social value. This scale is considered valid, parsimonious, reliable, and well-accepted [18,21,24]. The scale omits conditional and epistemic values as these are different values considered less critical to a general value measure [21]. Hence, several studies analyzed the product's perceived value under the three fundamental dimensions of functional, emotional, and social values [21,24,25]. This study agrees with this approach, considering these three dimensions to be a suitable framework for exploring the relationship between perceived value and pro-circular behaviors.

Numerous studies have explored the relationship between perceived value and sustainable considerations. For example, researchers investigated how consumers' perceptions of the value of environmental products influence their intentions and decisions to engage in pro-environmental behaviors (e.g., recycling, reducing energy consumption, choosing eco-friendly transportation options) [26–28]. Some studies examined how consumers perceive the value of eco-friendly products compared to conventional ones and the willingness to pay a premium for environmentally responsible products [29–32]. Another perspective analyzes the factors influencing consumers' valuation (e.g., attitudes, values, and environmental beliefs) [33]. Previous research also studies how the perceived value of environmentally friendly products can be affected by how consumers perceive the quality and performance of these products compared to non-sustainable alternatives [34].

Notably, the research on the relationship between perceived value and the circular economy explored how consumers perceive products designed for circularity, such as products made from recycled materials or items that are reused, refurbished, easily repairable, and recyclable [35–37]. This research venue often found consumers to be concerned with and possess low trust in these products [38]. Some research investigated how circular business models (e.g., product-as-a-service and take-back schemes) and products with circular attributes influence perceived value and consumer adoption of circular practices [39,40]. Finally, little research looks at whether the perceived value of contributing to a circular economy influences consumers' willingness to repair, recycle, share items with others, or resell products [41,42].

Although some research addressed the role of perceived value in the acceptance of circular business models [34,37,40] and the adoption of green products [26–28,34], most of this research focused on sustainable purchase intentions and behaviors. One possible reason is that the role of perceived value was oriented toward understanding the acquisition phase in the consumption stages. However, the new demands and needs for sustainable products and sustainable business models (e.g., circular business models) drive a more holistic perspective from marketing in which other behaviors beyond purchase must be analyzed to be promoted. To the best of our knowledge, little research analyzed how a construct primarily related to purchase, like perceived value, could impact the adoption of pro-circular behaviors. Moreover, the previous literature referred to the impact of circular attributes on perceived value [39]. However, the opposite direction has yet to be addressed: the effect of perceived value on circular behaviors.

2.2. Pro-Circular Behaviors, Perceived Value, and Green Consumption Values

Pro-circular behaviors can be defined as those consumer behaviors necessary in a circular economy [43]. These behaviors are framed in different Rs (e.g., reuse, recycle, reduce, recover, remanufacture, and redesign) [4]. From a sustainability perspective, procircular behaviors are linked to all consumption decisions (i.e., purchasing, product use, and final disposal) [43,44]. For instance, renting is a behavior that supports circular business models in the initial stage (i.e., purchase); meanwhile, repairing and reusing could be associated with the product-use stage. Resale and recycling, through return or devolution schemes, are most related to the end of the useful life that a product could have for a consumer (i.e., final disposal stage) [5].

Even though pro-circular behaviors in all stages of consumption are desirable, those practices focused on extending the lifespan of products were identified as crucial because of the potential positive environmental impact they could have on the whole value chain. Therefore, in this research, we consider pro-circular behaviors of repair and reuse and define these as follows: repair consists of putting back into operation a product that has ceased to serve its original purpose [11,45], and reuse consists of putting back into use a discarded product that is still in good condition and fulfills its original function [11,45].

We explore the role of perceived value on these pro-circular behaviors without modifying the functional, emotional, and social dimensions with environmental attributes. That is because we want to analyze how impact has a robust purchase-related construct, like perceived value on behaviors that, by definition, avoid the acquisition and consider extending the life cycle of a product. Nevertheless, previous research found the imperative role of environmental identity, concern, and attitude in adopting sustainable behaviors, including circular ones [21]. In fact, some authors pointed out that there could be other sources of value that consumers may obtain besides other types of value in a consumption experience [14]. Thus, we want to analyze the role of environmental values (a different source of value besides the product) as a moderator in the relationship between perceived value and pro-circular behaviors. We chose specific values focused on sustainable consumption, considering them closer to the purpose of this study. These values are defined under the construct of green consumption values.

The construct of green consumption values was introduced by Haws et al. [46], who defined it as "the tendency to express the value of environmental protection through one's purchases and consumption behaviors" (p. 2). Thus, consumption will be related to ethical choices that also consider the benefits for others (including the environment) besides the consumer's personal gains [47,48]. For example, consumers can avoid an early-end disposition (e.g., via reuse) or take disposal actions in their consumption focused on reducing waste and protecting natural resources (e.g., transferring the value of unused products to the next owners via resale) [42,46,49].

2.3. Hypotheses' Development

Previous research on perceived value highlighted the importance of the three dimensions (e.g., functional, emotional, and social) in purchase-related behaviors. For example, in the social commerce context, functional, emotional, and perceived social values significantly and positively affect purchase intention [50–57]. Also, all perceived value dimensions in luxury consumption were identified as critical in the consumer decision-making process [58,59]. Even in sustainability research, most of the studies suggested the importance of perceived value in green purchase intentions and behaviors [60–62].

Mainly, emotional value is a unique subjective emotion consumers feel in response to purchasing and using the product [10]. Therefore, previous research found that emotional value had positive effects directly on purchasing intentions or behavior [10,61] or indirectly on customer satisfaction [22,63], which increases the overall evaluation of the product/service [22] and, consequently, purchase and continuance intentions and behaviors [49,61,64,65]. Thus, feelings of joy and fun in consumption have a greater impact on consumers' choices [66]. Because of these findings, we reason that when people

experiment with positive feelings and emotions toward purchase behavior, they will not perform other behaviors that avoid shopping or the acquisition of products (e.g., repairing and reusing). Thus, we hypothesized that

H1. The emotional perceived value of the product will have a negative relationship with pro-circular behaviors of reusing and repairing.

The story is similar to perceived social value. Previous research pointed out how the more social value users obtain from using social commerce sites, purchasing a particular product (e.g., luxury items), or just acquiring branded goods, the more satisfaction and purchase intention they will develop [50,67,68]. Hence, we believe that when individuals appreciate the social consequences of purchase (e.g., individual's image, social status, recognition or approval from others), they do not want to lose those social benefits of acquiring products, so this perceived social value will have negative implications in practices that discourage purchasing, such as repairing or reusing items. Even consumers may worry that they cannot gain social approval because of certain features of circular products [10] (e.g., reused or repaired items), therefore obtaining low social value from these products. For instance, previous research on circular business models referred to the negative stigma attached to repair as a barrier to adopting this practice. Often, repaired products are associated with economic hardship and poverty, discouraging consumers from repairing and using repaired products (see Terzioğlu [7]). Hence, we propose the following hypothesis.

H2. The perceived social value of the product will have a negative relationship with pro-circular behaviors of reusing and repairing.

As mentioned, functional value positively affects purchase intentions and behaviors, even sustainable ones [60,61]. However, the scarce research on functional value and circular behaviors offers an additional perspective in which the product's utilitarian properties, features, and attributes could positively affect repairing and reusing behaviors. For instance, previous research on perceived value and repairing pointed out how functional value could motivate consumers to repair. The properties of products that fulfill the users' practical needs (i.e., functional attributes) are often important because some products are needed every day [7]. In addition, people could be discouraged from repairing products manufactured with low-quality materials or to last for a limited time; conversely, product endurance is considered a motivation to repair [7]. This idea fits with a great functional perceived value that covers good quality, endurance, good functional attributes, and an excellent cost–benefit relationship. So, we hypothesized that

H3. The functional perceived value of the product will have a positive relationship with the pro-circular behavior of repairing.

In addition, the previous literature mentioned that "if a product is functional, beautiful, and valuable, all at once, consumers will not want to throw it, but make optimal use of it" [69] (p. 4). Conversely, a barrier to reusing is the quality and durability of products and packaging [1]. Thus, some authors pointed out strategies such as increasing durability and improving care and maintenance as crucial for longer product lifetimes [70].

Under this reasoning, products' functional attributes could help used products retain value, motivating their reuse. So, we hypothesized that

H4. The functional perceived value of the product will have a positive relationship with the pro-circular behavior of reusing.

On the other hand, green consumption values are related to different constructs (e.g., environmental concerns). Previous research pointed out environmental concerns as a motivation to repair. It seems that the awareness of the damage that discarding a product

causes to the environment leads consumers to reduce such damage by repairing it [7] (p. 8). Thus, different studies found positive relationships between people's concerns about environmental issues and the frequency of repairing [41,71]. Based on these ideas, we proposed the following hypotheses:

H5. *The positive relationship between perceived functional value and repairing would be stronger* (*vs. weaker*) *for those consumers with high* (*vs. low*) *green consumption values.*

H6. *The negative relationship between perceived emotional/social values and repairing would be stronger (vs. weaker) for those consumers with low (vs. high) green consumption values.*

Similarly, the previous literature referred to consumers who reuse and choose reusable products as those influenced by environmental concerns about the consequences of waste [72], so their reusing behaviors are value-based [1]. Likewise, "previous studies have found that consumers with a high level of green consumption values are aware of their purchase practices that protect the environment through sustainable resale behavior" [42] (p. 4). These consumers take care of the products while using them, motivated to preserve the quality and then be able to resell those items, to be environmentally responsible consumers [73]. As reselling behavior aims to extend the lifetime of a product (like reusing behavior), we propose a similar effect of green consumption values on reusing behavior. In addition, some studies pointed out that consumers' environmental consciousness influenced reusing behaviors like donating clothing [74].

Thus, we hypothesized that

H7. *The positive relationship between perceived functional value and reusing would be stronger* (*vs. weaker*) *for those consumers with high* (*vs. low*) *green consumption values.*

H8. The negative relationship between perceived emotional/social values and reusing would be stronger (vs. weaker) for those consumers with low (vs. high) green consumption values.

3. Materials and Methods

3.1. Population and Sample

This research was developed in Medellín, Colombia. This city is recognized as the Latin American Capital of Electric Mobility [75,76], being an ecocity with intelligent urban equipment models, green spaces, and efficient waste management systems. There, we focused on the population of neighborhoods (Comunas) with a medium- to high-level quality of life (based on the Quality-of-Life Index (QLI)) [77]. These neighborhoods belong to the city's socioeconomic levels 3, 4, 5, and 6. The population in these neighborhoods was delimited by age (between 25 and 60 years old). A sample of this population was selected through stratified sampling according to the weight (percentage) of each neighborhood represented in the city population and applying the criteria and statistical developments determined by the literature [78].

Data were collected through a survey that covered several factors related to pro-circular consumption. This survey was programmed and distributed by a market research company to a sample of 417 people belonging proportionally to the neighborhoods previously defined. The panel of participants should have purchasing and consumption power over the products on which their behaviors would be evaluated (in this case, furniture and clothing). Of the sample, most respondents were female (59%), between 25 and 60 years of age (mean = 41 years). Most participants were single (37%), followed by married (34%), and 80.9% lived in socioeconomic strata 3 or 4. Likewise, most were employed (57%), followed by self-employed (30%), and 41% had a professional education level. Regarding their monthly income, 75% earned between COP 1,000,000 and COP 5,000,000; additionally, 53% lived in an apartment and 41% in a house. (See other sociodemographic details of the sample in the descriptive results for each behavior).

3.2. Variables and Measuring Instrument

3.2.1. Dependent Variables

As we mentioned before, we focused on two behaviors that aim to extend the useful life of products: repairing and reusing. These behaviors (our target variable) were measured using the scale proposed by Diddi and Yan [41]: this scale measures the frequency level with 5 points (1 never; 5 always). The specific items used to operationalize each behavior according to the products analyzed were "How often do you repair your furniture (e.g., chairs, tables, desks) instead of buying new ones?"; "How often do you repair your clothes instead of buying new ones; "How often do you reuse your furniture (e.g., desks); How often do you reuse your clothing".

3.2.2. Independent Variables

As previously explained in the theoretical framework, we explored the role of perceived value and its three dimensions (functional, emotional, and social values) as explanatory variables. We measured perceived value using the scale proposed by Saura and Vivó [79], involving 10 items (5 for functional value, 3 for emotional value, and 2 for social value). A Likert scale with four points measuring the level of agreement (1. Strongly disagree, 2. Disagree, 3. Agree, 4. Strongly agree). As shown in Appendix A, these items were adapted to the products evaluated.

3.2.3. Moderator Variable

We explored the role of green consumption values in the relationship between perceived value and pro-circular behaviors. This variable was measured using the scale proposed by Haws et al. [46], involving six items and a Likert scale with four levels of agreement (1. Strongly disagree, 2. Disagree, 3. Agree, 4. Strongly agree) (See Appendix A).

3.2.4. Control Variables

As control variables, we involved two specific factors that previous research pointed out as positively associated with the pro-circular behaviors under study (i.e., repairing and reusing) [5]: Moral Norms and Specific Perceived Consumer Effectiveness. Moral norms refer to the commitment to act correctly, representing an obligation the individual feels to behave consistently in favor of the environment [80]. This variable was measured using the scale proposed by Vining and Ebreo [81], which measures the level of agreement (1. Strongly disagree, 2. Disagree, 3. Agree, 4. Strongly agree) against three items ("I feel a strong obligation to…"; "I would feel guilty if…"; "I am willing to go the extra mile for…"). We adapted these items to the behaviors evaluated. On the other hand, perceived consumer effectiveness is the degree to which consumers believe that their actions can make a difference [82,83], achieving an environmental objective or solving an environmental problem [84]. We measured specific PCE for each behavior evaluated by adapting the scale proposed by Ellen et al. [82]. Appendix A shows the details of the items used to operationalize these variables.

In addition, as control variables, we considered the sociodemographic factors that were previously used in studies on pro-environmental and pro-circular behaviors (e.g., gender, age, marital status, socioeconomic status, occupation, educational level, income level, and type of housing).

In constructing variables, careful consideration was given to Cronbach's Alpha, the most commonly used criterion to evaluate the internal consistency of the items of a construct [85]. Table A1 shows that the lower Cronbach's Alpha values were from 0.6 to 0.7. These figures align with the threshold of acceptability, especially when accompanied by a corrected item-total correlation exceeding 0.3, as suggested by Ursachi et al. [85] and Raharjanti et al. [86]. Thus, the Cronbach's alpha values obtained demonstrated the reliability and internal consistency of the scales used [86,87]. (See Appendix A to see the details of the variables, items, and scales used, as well as Cronbach's alpha for each variable)

The Ordinal Logit model was implemented to address the relationship between predictor or independent variables and the ordinal scale of the response variable, recognized for its suitability in cases where the dependent variable has multiple ordered levels. The Ordinal Logit model is also known as the proportional odds model, and another popular option is known as the stereotype model. Ordered logit models were derived, starting with a binary logit model and generalizing it to allow for more than two outcomes [87].

Four models were specified to examine the relationship, with the dependent variable being repairing and reusing behaviors associated with furniture and clothing products. These variables are ordinal with five categories, where the lowest category represents "Never" and the highest "Always". Independent variables include perceived values for each product from a functional, emotional, and social perspective. The moderating variable is the Green Consumption Value Index, and control variables are associated with the norms index for each behavior and product, the Specific Perceived Consumer Effectiveness (SPCE) associated with each behavior, and sociodemographic variables such as age, gender, social stratum, income level, education level, occupation, marital status, and type of housing.

Additionally, quantitative variables (perceived values, values index, and norms index) were normalized to allow for comparison and interpretation, as their values are constrained between 0 and 1. The stepwise methodology followed by Nordberg [88] and Soroush et al. [89] was implemented to determine significant control variables in the model. This methodology considers the correlation between variables and helps select those that significantly contribute to explaining the dependent variable.

For the four behaviors analyzed, two models were estimated to deepen the understanding of the dynamics between independent variables associated with perceived values. In the first model, the Green Consumption Value Index is incorporated as a control variable to fine-tune or nullify its effects on the relationship between the independent variables of perceived values and the dependent variable of behaviors. This inclusion aims to mitigate any potential confounding influences arising from the Green Consumption Value Index, thereby refining the examination of how the perceived values of individuals correlate with their behavioral patterns. In the second model, the green consumption value index was included as a moderating variable and its interaction with perceived values, following the methodology proposed by Saunders [90], Sharma et al. [91], and Dawson [92]. This approach explored its role in the relationship between predictor variables and the ordinal response through a two-way interaction. This approach allowed a comprehensive and differentiated evaluation of the impact of the variables of interest on the studied phenomenon by calculating probabilities for the "Always" response category.

4. Results

4.1. Descriptive Statistics

We summarized the main descriptive and inferential findings associated with quantitative variables (e.g., Perceived Value Index, Green Consumption Value Index, and Moral Standards Index, which take values between 1 and 4 based on their response scale) and qualitative variables (e.g., Specific Perceived Consumer Effectiveness and some sociodemographic variables) for each behavior. These findings are based on calculating frequencies, column percentages, mean, and standard deviations for each behavior. The means are presented for quantitative variables, and their standard deviations are in parentheses; for qualitative variables, the frequency and column percentage are in parentheses.

The likelihood ratio test was used to analyze the association between qualitative independent variables and response variables associated with the behaviors of each product. The linear regression test was also employed to analyze the difference between the means of k samples for quantitative variables. The P-value of these tests is presented in the last column of the tables. The control variables that were not statistically significant from the univariate analysis will only be removed from the ordinal logistic regression model once the multivariate validation process by steps is completed.

All the behaviors analyzed present similar descriptive results. People repair and reuse occasionally and often; functional perceived value exhibits the highest mean, and green consumption values increase as people reuse and repair with more frequency (both categories), suggesting the importance of this variable to behave in a pro-circular way. Moreover, environmental values, the perception of the effectiveness of repairing and reusing behaviors to solve environmental issues, and the personal feeling of doing the right thing when repairing and reusing are essential variables affecting people's pro-circular behaviors that extend the lifespan of furniture and clothing products. We detailed the descriptive statistics for each behavior below.

4.1.1. Repair Furniture

We found that most people in the sample repair furniture occasionally (39.33%) and often (32.37%). Table 1 shows that the functional value has the highest average (3.16), and the social value has the lowest (2.19). Regarding the Green Consumption Value Index, it is observed that the average increases as one moves up the scale, going from 2.94 for the "never" option to 3.45 for the "always" option. This result suggests that consumers with more environmental values towards consumption repair with more frequency (see Table 1).

Table 1. Descriptive statistics (significant results for repairing behavior; furniture category).

How Often Do You Repair Your Furniture (e.g., Chairs, Tables, Desks) Instead of Buying New Ones?							
	Never	Rarely	Occasionally	Often	Always	Total	Test (<i>p</i> -Value)
n	18.00 (4.32%)	51.00 (12.23%)	164.00 (39.33%)	135.00 (32.37%)	49.00 (11.75%)	417.00 (100.00%)	
Emotional Value	3.20 (0.69)	2.83 (0.53)	2.78 (0.54)	2.68 (0.62)	2.69 (0.78)	2.76 (0.61	0.01
Functional Value	2.99 (0.54)	3.06 (0.48)	3.09 (0.32)	3.19 (0.39)	3.49 (0.37)	3.16 (0.40)	<0.00
Social Value	2.03 (0.83)	2.23 (0.64)	2.18 (0.72)	2.18 (0.78)	2.28 (1.08)	2.19 (0.78)	0.82
Green Values	2.94 (0.64)	3.05 (0.47)	3.11 (0.50)	3.29 (0.50)	3.45 (0.50)	3.19 (0.52)	<0.00
Moral Norm	2.22 (0.63)	2.58 (0.62)	2.83 (0.43)	3.12 (0.44)	3.52 (0.47)	2.95 (0.56)	<0.00
SPCE							
I strongly disagree	2.00 (11.11%)	1.00 (1.96%)	3.00 (1.83%)	2.00 (1.48%)	1.00 (2.04%)	9.00 (2.16%)	< 0.00
I disagree	7.00 (38.89%)	7.00 (13.73%)	19.00 (11.59%)	4.00 (2.96%)	2.00 (4.08%)	39.00 (9.35%)	
I agree	8.00 (44.44%)	36.00 (70.59%)	115.00 (70.12%)	94.00 (69.63%)	12.00 (24.49%)	265.00 (63.55%)	
I strongly agree	1.00 (5.56%)	7.00 (13.73%)	27.00 (16.46%)	35.00 (25.93%)	34.00 (69.39%)	104.00 (24.94%)	
Marital Status							
Single	6.00 (33.33%)	26.00 (50.98%)	59.00 (35.98%)	46.00 (34.07%)	19.00 (38.78%)	156.00 (37.41%)	0.06
Married	7.00 (38.89%)	11.00 (21.57%)	56.00 (34.15%)	49.00 (36.30%)	20.00 (40.82%)	143.00 (34.29%)	
Free union	4.00 (22.22%)	12.00 (23.53%)	34.00 (20.73%)	26.00 (19.26%)	10.00 (20.41%)	86.00 (20.62%)	
Widowed	1.00 (5.56%)	0.00 (0.00%)	2.00 (1.22%)	0.00 (0.00%)	0.00 (0.00%)	3.00 (0.72%)	
Other	0.00 (0.00%)	2.00 (3.92%)	13.00 (7.93%)	14.00 (10.37%)	0.00 (0.00%)	29.00 (6.95%)	
Level of studies							
Primary	0.00 (0.00%)	0.00 (0.00%)	1.00 (0.61%)	0.00 (0.00%)	0.00 (0.00%)	1.00 (0.24%)	0.06
Secondary	1.00 (5.56%)	2.00 (3.92%)	5.00 (3.05%)	7.00 (5.19%)	3.00 (6.12%)	18.00 (4.32%	
Technical	0.00 (0.00%)	2.00 (3.92%)	24.00 (14.63%)	13.00 (9.63%)	5.00 (10.20%)	44.00 (10.55%)	
Technological	4.00 (22.22%)	7.00 (13.73%)	30.00 (18.29%)	19.00 (14.07%)	5.00 (10.20%)	65.00 (15.59%)	
Professional	4.00 (22.22%)	26.00 (50.98%)	70.00 (42.68%)	50.00 (37.04%)	20.00 (40.82%)	170.00 (40.77%)	
Specialization	8.00 (44.44%)	9.00 (17.65%)	26.00 (15.85%)	33.00 (24.44%)	16.00 (32.65%)	92.00 (22.06%)	
Master's Degree	1.00 (5.56%)	5.00 (9.80%)	6.00 (3.66%)	10.00 (7.41%)	0.00 (0.00%)	22.00 (5.28%)	
PhD	0.00 (0.00%)	0.00 (0.00%)	2.00 (1.22%)	3.00 (2.22%)	0.00 (0.00%)	5.00 (1.20%)	
Type of housing							
House	5.00 (27.78%)	20.00 (39.22%)	69.00 (42.07%)	49.00 (36.30%)	29.00 (59.18%)	172.00 (41.25%)	0.05
Housing complex	1.00 (5.56%)	0.00 (0.00%)	6.00 (3.66%)	6.00 (4.44%)	2.00 (4.08%)	15.00 (3.60%)	
Apartment-complex	12.00 (66.67%)	31.00 (60.78%)	88.00 (53.66%)	75.00 (55.56%)	18.00 (36.73%)	224.00 (53.72%)	
Rural housing	0.00 (0.00%)	0.00 (0.00%)	1.00 (0.61%)	5.00 (3.70%)	0.00 (0.00%)	6.00 (1.44%)	

Note: **SPCE: Specific Perceived Consumer Effectiveness:** *By repairing my furniture (e.g., tables, chairs, desks), I can contribute to solving environmental problems.* Significant *p*-values in **bold.** Linear regression test performed for quantitative variables. Likelihood ratio test for qualitative variables.

From the likelihood ratio test, we observe that the covariables Specific Perceived Consumer Effectiveness—SPCE, marital status, education level, and type of housing showed a significant association. As for the linear regression analysis, we observe that Perceived Emotional Value and Perceived Functional Value and the control variables of Green Consumption Values and Moral Norms (feeling of moral obligation to repair furniture) showed a significant mean difference. This indicates that they are essential variables affecting people's behavior when repairing furniture.

4.1.2. Repair Clothing

As Table 2 shows, most people repair clothing occasionally (33.09%) and often (29.98%). Like furniture repairing behavior, the functional value has the highest average (3.16), and the social value has the lowest (2.42). Regarding the Green Consumption Value Index, it is observed that the average increases as one moves up the scale, going from 2.84 for the "never" option to 3.61 for the "always" option. Thus, the more individuals exhibit environmental values towards consumption, the more they frequently repair their clothes (see Table 2).

Table 2. Descriptive statistics (significant results for repairing behavior; clothing category).

		How Of	ten Do You Repair	r Your Clothes Ins	tead of Buying N	ew Ones?	
	Never	Rarely	Occasionally	Often	Always	Total	Test (<i>p</i> -Value)
n	33.00 (7.91%)	63.00 (15.11%)	138.00 (33.09%)	125.00 (29.98%)	58.00 (13.91%)	417.00 (100.00%)	
Emotional Value	3.38 (0.51)	3.18 (0.47)	3.05 (0.46)	2.88 (0.56)	2.83 (0.73)	3.02 (0.56)	<0.00
Functional Value	3.22 (0.46)	3.08 (0.33)	3.11 (0.35)	3.12 (0.38)	3.46 (0.43)	3.16 (0.40)	< 0.00
Social Value	2.68 (0.91)	2.44 (0.72)	2.43 (0.81)	2.38 (0.79)	2.33 (1.08)	2.42 (0.84)	0.38
Green Values	2.84 (0.64)	3.08 (0.44)	3.09 (0.53)	3.26 (0.42)	3.61 (0.39)	3.19 (0.52)	<0.00
Moral Norm	2.04 (0.68)	2.44 (0.58)	2.71 (0.44)	3.07 (0.45)	3.51 (0.45)	2.84 (0.63)	<0.00
SPCE							
I strongly disagree	8.00 (24.24%)	2.00 (3.17%)	2.00 (1.45%)	1.00 (0.80%)	1.00 (1.72%)	14.00 (3.36%)	< 0.00
I disagree	14.00 (42.42%)	21.00 (33.33%)	24.00 (17.39%)	6.00 (4.80%)	2.00 (3.45%)	67.00 (16.07%)	
I agree	9.00 (27.27%)	33.00 (52.38%)	92.00 (66.67%)	86.00 (68.80%)	18.00 (31.03%)	238.00 (57.07%)	
I strongly agree	2.00 (6.06%)	7.00 (11.11%)	20.00 (14.49%)	32.00 (25.60%)	37.00 (63.79%)	98.00 (23.50%)	
Gender							
Female	21.00 (63.64%)	42.00 (66.67%)	65.00 (47.10%)	83.00 (66.40%)	35.00 (60.34%)	246.00 (58.99%)	0.01
Male	12.00 (36.36%)	21.00 (33.33%)	73.00 (52.90%)	42.00 (33.60%)	23.00 (39.66%)	171.00 (41.01%)	

Note: **SPCE: Specific Perceived Consumer Effectiveness:** *By repairing my clothes instead of buying new ones, I can contribute to solving environmental problems.* Significant *p*-values in **bold.** Linear regression test performed for quantitative variables. Likelihood ratio test for qualitative variables.

In this case, we observe that Specific Perceived Consumer Effectiveness—SPCE and gender showed a significant association with clothing repairing behavior. Based on the linear regression analysis, we also observe that Perceived Emotional Value and Perceived Functional Value, as explanatory variables, and Green Consumption Values and Moral Norms, as control variables, showed a significant mean difference (see Table 2). This finding suggests that these variables significantly affect people's behavior when repairing clothes.

4.1.3. Reuse Furniture

Our results show that most people in the sample reuse their furniture often (38.61%) and occasionally (32.37%). As Table 3 shows, the functional value has the highest average (3.16), followed by the emotional value (2.76), and the social value with the lowest average (2.19). The average of the Green Consumption Values Index increases as one moves up the scale, going from 2.67 for the "never" option to 3.41 for the "always" option. In the same way as repairing, this finding suggests that consumers with more environmental values towards consumption are those who reuse their furniture more frequently (see Table 3).

		How Of	ten Do You Reuse	Your Furniture (e.	.g., Chairs, Tables	, Desks)?	
	Never	Rarely	Occasionally	Often	Always	Total	Test (<i>p</i> -Value)
n	13.00 (3.12%)	40.00 (9.59%)	135.00 (32.37%)	161.00 (38.61%)	68.00 (16.31%)	417.00 (100.00%)	
Emotional Value Functional Value Social Value	3.13 (0.74) 2.98 (0.51) 2.15 (0.85)	2.84 (0.49) 3.04 (0.44) 2.21 (0.61)	2.83 (0.55) 3.08 (0.35) 2.19 (0.76)	2.65 (0.55) 3.13 (0.37) 2.19 (0.75)	2.77 (0.83) 3.48 (0.38) 2.18 (0.97)	2.76 (0.61) 3.16 (0.40) 2.19 (0.78)	0.01 < 0.00 1.00
Green Values	2.67 (0.57)	3.03 (0.41)	3.13 (0.51)	3.24 (0.49)	3.41 (0.53)	3.19 (0.52)	<0.00
Moral Norm	2.21 (0.67)	2.60 (0.43)	2.97 (0.34)	3.19 (0.44)	3.65 (0.39)	3.11 (0.52)	<0.00
SPCE							
I strongly disagree	2.00 (15.38%)	1.00 (2.50%)	0.00 (0.00%)	1.00 (0.62%)	1.00 (1.47%)	5.00 (1.20%)	< 0.00
I disagree	1.00 (7.69%)	4.00 (10.00%)	10.00 (7.41%)	6.00 (3.73%)	1.00 (1.47%)	22.00 (5.28%)	
I agree	8.00 (61.54%)	28.00 (70.00%)	91.00 (67.41%)	82.00 (50.93%)	23.00 (33.82%)	232.00 (55.64%)	
I strongly agree	2.00 (15.38%)	7.00 (17.50%)	34.00 (25.19%)	72.00 (44.72%)	43.00 (63.24%)	158.00 (37.89%)	
Marital Status							
Single	4.00 (30.77%)	20.00 (50.00%)	57.00 (42.22%)	55.00 (34.16%)	20.00 (29.41%)	156.00 (37.41%)	0.05
Married	4.00 (30.77%)	6.00 (15.00%)	48.00 (35.56%)	55.00 (34.16%)	30.00 (44.12%)	143.00 (34.29%)	
Free union	3.00 (23.08%)	12.00 (30.00%)	20.00 (14.81%)	39.00 (24.22%)	12.00 (17.65%)	86.00 (20.62%)	
Widowed	1.00 (7.69%)	1.00 (2.50%)	1.00 (0.74%)	0.00 (0.00%)	0.00 (0.00%)	3.00 (0.72%)	
Other	1.00 (7.69%)	1.00 (2.50%)	9.00 (6.67%)	12.00 (7.45%)	6.00 (8.82%)	29.00 (6.95%)	

Table 3. Descriptive statistics (significant results for reusing behavior; furniture category).

Note: **SPCE: Specific Perceived Consumer Effectiveness:** *By reusing my furniture (e.g., tables, chairs, desks), I can contribute to solving environmental problems.* Significant *p*-values in **bold.** Linear regression test performed for quantitative variables. Likelihood ratio test for qualitative variables.

Table 3 also shows that, from the likelihood ratio test, we observe that the covariables Specific Perceived Consumer Effectiveness—SPCE and marital status showed a significant association. As for the linear regression analysis, we observe that Perceived Emotional Value and Perceived Functional Value, as well as the control variables of Green Consumption Values and Moral Norms (feeling of moral obligation to reuse furniture), showed a significant mean difference. This finding indicates that they are essential variables affecting people's behavior when reusing furniture.

4.1.4. Reuse Clothing

We found that most people in the sample reuse clothing often (35.01%) and occasionally (28.06%). The functional value has the highest average (3.16), followed by the emotional value (3.02) and the social value (2.42) (see Table 4). Regarding the Green Consumption Value Index, it is observed that the average increases as one moves up the scale, going from 2.97 for the "never" option to 3.42 for the "always" option (see Table 4). Hence, the more consumers exhibit environmental values towards consumption, the more they frequently reuse their clothes.

Just as with repairing behavior, we observe that Specific Perceived Consumer Effectiveness—SPCE and gender showed a significant association with clothing-reusing behavior. We also observe that Perceived Emotional Value and Perceived Functional Value, as well as the control variables of Green Consumption Values and Moral Norms, showed a significant mean difference (see Table 4), indicating that they are essential variables affecting people's behavior when reusing clothing.

4.2. Hypotheses Testing

As explained in the methodology section, we implemented an Ordinal Logit model to test our hypotheses. We specified four models to examine the relationship between the three dimensions of perceived value (independent variables) and the repairing and reusing behaviors associated with furniture and clothing products (dependent variables). For the four behaviors, we estimated two models: In the first model, the Green Consumption Value Index was incorporated as a control variable to nullify its effects on the relationship between the independent variables of perceived values and the dependent variable of behaviors. In the second model, the Green Consumption Value index was included as a moderating variable. This model tests the interaction between this variable and perceived values.

Table 4. Descriptive statistics (significant results for reusing behavior; clothing category).

	How Often Do You Repair Your Clothes Instead of Buying New Ones?						
	Never	Rarely	Occasionally	Often	Always	Total	Test (<i>p</i> -Value)
n	32.00 (7.67%)	47.00 (11.27%)	117.00 (28.06%)	146.00 (35.01%)	75.00 (17.99%)	417.00 (100.00%)	
Emotional Value Functional Value Social Value	3.32 (0.54) 3.14 (0.46) 2.58 (0.91)	3.12 (0.50) 3.09 (0.43) 2.45 (0.69)	2.98 (0.49) 3.15 (0.32) 2.38 (0.80)	2.97 (0.55) 3.09 (0.37) 2.41 (0.82)	2.95 (0.67) 3.38 (0.43) 2.44 (1.01)	3.02 (0.56) 3.16 (0.40) 2.42 (0.84)	0.01 < 0.00 0.82
Green Values	2.97 (0.58)	3.07 (0.42)	3.10 (0.50)	3.24 (0.55)	3.42 (0.42)	3.19 (0.52)	<0.00
Moral Norm	2.20 (0.66)	2.65 (0.46)	2.86 (0.47)	3.21 (0.45)	3.56 (0.39)	3.04 (0.60)	<0.00
SPCE							
I strongly disagree I disagree I agree I strongly agree	6.00 (18.75%) 10.00 (31.25%) 16.00 (50.00%) 0.00 (0.00%)	0.00 (0.00%) 13.00 (27.66%) 32.00 (68.09%) 2.00 (4.26%)	4.00 (3.42%) 16.00 (13.68%) 83.00 (70.94%) 14.00 (11.97%)	1.00 (0.68%) 11.00 (7.53%) 89.00 (60.96%) 45.00 (30.82%)	1.00 (1.33%) 2.00 (2.67%) 24.00 (32.00%) 48.00 (64.00%)	12.00 (2.88%) 52.00 (12.47%) 244.00 (58.51%) 109.00 (26.14%)	<0.00
Gender Female Male	19.00 (59.38%) 13.00 (40.62%)	28.00 (59.57%) 19.00 (40.43%)	58.00 (49.57%) 59.00 (50.43%)	86.00 (58.90%) 60.00 (41.10%)	55.00 (73.33%) 20.00 (26.67%)	246.00 (58.99%) 171.00 (41.01%)	0.03

Note: **SPCE: Specific Perceived Consumer Effectiveness:** *By reusing my clothes, I can contribute to solving environmental problems.* Significant *p*-values in **bold.** Linear regression test performed for quantitative variables. Likelihood ratio test for qualitative variables.

Our first hypothesis was that the emotional perceived value of the product would have a negative relationship with reusing and repairing behaviors. The results support this hypothesis just for clothing repairing behavior. As Table 5 shows in Model 1, there is a negative and significant relationship between emotional value and clothing repairing behavior (Coef = -1.59; p < 0.01). Thus, as the perceived emotional value of the product increases, keeping the other variables constant, the probability of it being repaired decreases. We also stated a negative relationship between social value and repairing/reusing behaviors. However, this second hypothesis was not supported for any of the behaviors.

We found a positive relationship between functional value and repairing behaviors (Clothing: Coeff = 2.15; p < 0.01; Furniture: Coef = 3.08; p < 0.01) (See Tables 5 and 6). This supports H3. Hence, as the perceived functional value of the products (furniture and clothing) increases, keeping the other variables constant, the probability of them being repaired is higher. Functional value also has a positive effect on reusing (H4), but just for the furniture category (Coef = 1.74; p < 0.05), as Table 7 shows.

Regarding the moderator role of green consumption values, we found the following: The interaction between functional perceived value and green consumption values was positive and significant for clothing repairing behavior (Coef = 9.09; p < 0.05), supporting the H5 for the clothing category (See Table 5, Model 2). As Table 5 shows, when green consumption values are taken as a moderating variable, the effect between perceived functional value and clothing repair behavior intensifies, especially for those with a high perceived functional value for their product (see Probability Outcomes for Functional Value). Thus, as the perceived functional value increases, the probability of always repairing also increases, even more so when green consumption values are present. The same occurs for reusing behavior for both categories. As we can see in Models 2 of Tables 7 and 8, the interaction between functional value and green consumption values was positive and significant for reusing behavior (Furniture: Coef = 7.40; p < 0.1; Clothing: Coef = 7.86; p < 0.05). This supports H7. Hence, when green consumption values are used as a moderating variable, the effect between perceived functional value and the reusing furniture/clothing behavior intensifies, especially for those with a high perceived functional value for their product (see Probability Outcomes in Tables 7 and 8, respectively). Thus, as

the perceived functional value increases, the probability of always reusing also increases, even more so when green consumption values are present.

Table 5. Hypotheses testing for repairing behavior: clothing category.

			Model 1			Model 2	
Variables	Categories	Estimate	S.E	OR	Estimate	S.E	OR
Perceived Emotional Value Index		-1.59 ***	0.60	0.20	1.73	2.19	5.63
Perceived Functional Value Index		2.15 ***	0.79	8.55	-4.09	2.86	0.02
Perceived Social Value Index		0.14	0.37	1.16	-3.67 **	1.48	0.03
Moral Norm Index		5.90 ***	0.66	364.25	5.89 ***	0.66	361.14
Specific Perceived Consumer Effectiveness	I disagree	0.62	0.66	1.86	0.87	0.67	2.39
-	I agree	1.24 *	0.66	3.45	1.52 **	0.66	4.57
	I strongly agree	1.60 **	0.69	4.94	1.89 ***	0.71	6.64
Green Consumption Value Index		1.81 ***	0.59	6.11	-4.40	3.26	0.01
Emotional Value *Green Consump Values					-4.56	2.89	0.01
Functional Value *Green Consump Values					9.09 **	3.96	8828.08
Social Value *Green Consump Value					5.35 ***	1.99	210.85
	Observations	417			417		
	LR chi2	257.51			267.99		
	Prob > chi2	0			0		
	McFadden's r2	0.21			0.22		
	Green Consumpti	on Values Mo	deration d	etails:			
Scenario			Model 1			Model 2	
	Probability Outco (Functional Value	mes of the "A "Green Consu	lways″ res umption Va	ponse alues)			
			-		Green Consu	mption Valu	es Index = 1
Perceived Functional Value Index = 0			0.042 *			0.009	
Perceived Functional Value Index = 0.25			0.065 ***			0.028	
Perceived Functional Value Index = 0.5			0.098 ***			0.079 ***	
Perceived Functional Value Index = 0.75			0.141 ***			0.189 ***	
Perceived Functional Value Index = 1			0.198 ***			0.365 ***	
	Probability Outco (Social Value*C	mes of the "A Green Consum	lways" res	ponse les)			
					Green Consu	mption Valu	es Index = 1
Perceived Social Value Index = 0			0.135 ***			0.114 ***	
Perceived Social Value Index = 0.25			0.138 ***			0.147 ***	
Perceived Social Value Index = 0.5			0.141 ***			0.187 ***	
Perceived Social Value Index = 0.75			0.144 ***			0.234 ***	
Perceived Social Value Index = 1			0.147 ***			0.288 ***	
					Green Consu	mption Valu	ies Index = 0
Perceived Social Value Index = 0			0.135 ***			0.168 *	
Perceived Social Value Index = 0.25			0.138 ***			0.088 **	
Perceived Social Value Index = 0.5			0.141 ***			0.042 **	
Perceived Social Value Index = 0.75			0.144 ***			0.019 *	

*** p < 0.01, ** p < 0.05, * p < 0.1.

We found a significant interaction between perceived social value and green consumption values for repairing behavior, at least regarding the clothing category (Coef = 5.35; p < 0.01) (See Model 2 in Table 5). Here, the role of green consumption values is relevant. As Table 5 shows, when green consumption values are present, the effect between perceived social value and the behavior of repairing becomes significant and intensifies. Thus, as the perceived social value increases, the probability of always repairing also increases, even more so when green consumption values are present. For those who do not have green consumption values, the probability of always repairing decreases considerably as the perceived social value of the product increases (see Probability Outcomes for Social Value in Table 5). Because we did not find evidence for repairing in the furniture category, nor for emotional value in any category, these results partially support H6. Finally, there were no significant interactions between emotional/social values and reusing behavior, so H8 was unsupported.

			Model 1			Model 2	
Variables	Categories	Estimate	S.E	OR	Estimate	S.E	OR
Perceived Emotional Value Index		-0.65	0.52	0.52	1.10	2.20	3.02
Perceived Functional Value Index		3.08 ***	0.83	21.73	-1.22	3.16	0.30
Perceived Social Value Index		0.63	0.41	1.87	0.98	1.53	2.66
Moral Norm Index		5.96 ***	0.65	386.34	5.91 ***	0.66	369.70
Socioeconomic status	4	0.05	0.21	1.05	0.02	0.21	1.02
	5	-0.31	0.28	0.73	-0.33	0.28	0.72
	6	1.17 ***	0.45	3.24	1.15 **	0.45	3.17
Green Consumption Value Index		0.92	0.58	2.52	-1.58	3.28	0.21
Emotional Value * Green Consump Values					-2.25	2.88	0.10
Functional Value * Green Consump Values					5.70	4.16	297.98
Social Value * Green Consump Values					-0.44	2.05	0.65
	Observations	417			417		
	LR chi2	174.47			177.13		
	Prob > chi2	0			0		
	McFadden's r2	0.15			0.15		

Table 6. Hypotheses testing for repairing behavior: furniture category.

*** p < 0.01, ** p < 0.05, * p < 0.1.

 Table 7. Hypotheses testing for reusing behavior: furniture category.

			Model 1			Model 2	
Variables	Categories	Estimate	S.E	OR	Estimate	S.E	OR
Perceived Emotional Value Index		-0.42	0.52	0.66	1.05	2.26	2.85
Perceived Functional Value Index		1.74 **	0.83	5.72	-3.78	3.25	0.02
Perceived Social Value Index		0.11	0.41	1.11	0.37	1.57	1.45
Moral Norm Index		8.25 ***	0.78	3830.28	8.19 ***	0.78	3603.84
Marital Status	Married	0.31	0.22	1.36	0.33	0.23	1.40
	Free union	0.26	0.26	1.30	0.27	0.26	1.32
	Widowed	-1.94 *	1.07	0.14	-1.86 *	1.06	0.16
	Other	0.68 *	0.38	1.98	0.65 *	0.38	1.92
Green Consumption Value Index		0.50	0.59	1.64	-3.54	3.40	0.03
Emotional Value * Green Consump Values					-1.87	2.98	0.15
Functional Value * Green Consump Values					7.40 *	4.27	1632.13
Social Value * Green Consump Values					-0.29	2.09	0.74
	Observations	417			417		
	LR chi2	215.13			218.73		
	Prob > chi2	0			0		
	McFadden's r2	0.19			0.19		
	Green Consumpti	ion Values M	oderation d	etails:			
Scenario			Model 1			Model 2	
	Probability Outco	mes of the "A	Always" res	ponse			
	(Functional Value	e*Green Cons	umption Va	lues)			
					Green Consu	mption Valu	ue Index = 1
Perceived Functional Value Index = 0			0.062 **			0.020	
Perceived Functional Value Index = 0.25			0.087 ***			0.046	
Perceived Functional Value Index = 0.5			0.121 ***			0.095 ***	
Perceived Functional Value Index = 0.75			0.162 ***			0.179 ***	
Perceived Functional Value Index = 1			0.212 ***			0.299 ***	

*** p < 0.01, ** p < 0.05, * p < 0.1.

Table 8. Hypotheses testing for reusing behavior: clothing category.

		Model 1			Model 2			
Variables	Categories	Estimate	S.E	OR	Estimate	S.E	OR	
Perceived Emotional Value Index		-0.86	0.58	0.42	-1.09	2.21	0.34	
Perceived Functional Value Index		0.75	0.79	2.11	-4.69 *	2.74	0.01	
Perceived Social Value Index		0.11	0.38	1.12	-1.63	1.53	0.20	
Moral Norm Index		6.19 ***	0.69	487.91	6.27 ***	0.70	529.34	
Gender	Male	-0.29	0.19	0.75	-0.26	0.19	0.77	
Specific Perceived Consumer Effectiveness	I disagree	0.86	0.70	2.35	0.85	0.70	2.35	

Table 8. Cont.

			Model 1			Model 2	
Variables	Categories	Estimate	S.E	OR	Estimate	S.E	OR
	I agree	0.86	0.67	2.36	0.89	0.67	2.43
	I strongly agree	1.79 **	0.70	5.99	1.80 **	0.70	6.07
Green Consumption Value Index		0.57	0.58	1.77	-6.61 **	3.10	0.00
Emotional Value Index * Green Consump Values					0.37	2.87	1.45
Functional Value Index * Green Consump Values					7.86 **	3.72	2581.70
Social Value Index * Green Consump Values					2.47	2.04	11.86
	Observations	417			417		
	LR chi2	227.21			232.89		
	Prob > chi2	0			0		
	McFadden´s r2	0.18			0.19		
Gı	een Consumption V	Values Moder	ation details	:			
Scenario			Model 1			Model 2	
Pro (Fu	obability Outcomes inctional Value * Gi	of the "Alwa reen Consum	iys" respons ption Values	e ;)			
					Gree	n Consum _l	otion
					Va	lue Index =	:1
Perceived Functional Value Index = 0			0.125 **			0.033	
Perceived Functional Value Index = 0.25			0.142 ***			0.065 **	
Perceived Functional Value Index = 0.5			0.160 ***			0.118 ***	
Perceived Functional Value Index = 0.75			0.179 ***			0.197 ***	
Perceived Functional Value Index = 1			0.200 ***			0.301 ***	

*** p < 0.01, ** p < 0.05, * p < 0.1.

5. Discussion

This research aimed to explore the relationship between the three dimensions of perceived value (e.g., functional, emotional, and social) and pro-circular behaviors of repairing and reusing, as well as the role of green consumption values in this relationship. We operationalized repairing and reusing behaviors in two product categories: clothing and furniture. Most marketing and circular economy studies focused on analyzing perceived value's role in the acquisition or purchase phase. Thus, this study contributes to a more holistic perspective by proposing an analysis oriented to understanding how perceived value could influence the adoption of other behaviors focused on the products' life-span extension, such as repair and reuse. This approach is worth it, not only to add to the few studies focused on the consumer as an essential actor in the circular economy but also to guide future research and marketing strategies toward promoting circular business models as one of the clear paths to tackle environmental and social issues. Ordinal logit models were implemented to test our hypotheses.

We proposed that emotional and perceived social values were negatively related to pro-circular behaviors of repairing and reusing. The hypothesis on emotional value was supported for the clothing category in repairing behavior. Thus, the emotional value of buying clothes affected consumers' clothing repairing behaviors negatively. Previous studies indicate that the likelihood of repairing clothes increases when people have a personal connection with such clothes [93,94]. In addition, the consumers' experiential relationship with clothing during its use phase determines how clothing is cared for, maintained, and repaired [41]. Hence, the attachment and love to clothes as a way of emotional value could encourage repairing behaviors [71,93]. These findings indicate that such emotional value should be oriented to highlight positive feelings during the use phase. Otherwise, as our results suggest, focusing on the emotional value of buying clothes will be in contrast to clothing repairing behaviors. We did not find support regarding emotional value and clothing reusing behavior. It seems that positive feelings involved in buying clothes (e.g., joy, liking) have nothing to do with wanting to extend the life span of this kind of product via reusing. Some studies find consumers enjoying shopping experiences related to pro-circular behaviors like buying secondhand clothing [95]; more research on

these practices will be needed to understand the relationship between emotional perceived value and reusing behaviors.

Results about furniture were different, and the corresponding hypotheses were not supported. Little research on furniture pointed out that life stages (e.g., marriage, parenthood, home purchase, career advancement) trigger furniture purchases [13], so people were expected to enjoy furniture shopping and prefer this behavior instead of repairing or reusing those items. Nevertheless, it seems that the emotional value of buying furniture has nothing to do with wanting to extend the life span of these products via repairing and reusing. Probably, because our sample had a median age of around 40, and "household furniture consumption involves a much longer time horizon than many other consumer goods" [13] (p. 26), they have not had the opportunity to see aging their furniture, so reusing/repairing decisions were not needed until now. However, as we suggested for clothing, emotional value focused on positive feelings and emotions during the use phase could also be enhanced for the furniture category. As research pointed out, furniture purchases could be driven by consumers' emotional needs [13]. Furniture pieces are valued by the memories of people, occasions, and relationships those items aroused [96], even becoming an extension of the self [97]; fostering this emotional perceived value during the use of furniture could be helpful to promote business models oriented to repairing and reusing behaviors.

Regarding the social value, our findings do not support the hypotheses proposed. The current environmental and social issues changed the mindset of several people concerned about these problems. The image people want to show to others may be more pro-social and eco-friendly. So, even though social approval is important for consumers, this approval may change to another kind of behavior (e.g., green purchase and donating behaviors [98]). In fact, social norms are one of the main factors to act pro-environmentally [99,100]. However, it seems that the social benefits of purchase are not related yet to pro-circular behaviors. As our results show, social image is not related to repairing and reusing practices. Because most studies on perceived value and sustainable behaviors were focused on the initial consumption stage (i.e., purchase), little information can support what happens to sustainable behaviors during use and disposal. Our study gives an initial idea that perceived social value has nothing to do with repairing and reusing behaviors. More research is needed to understand how social benefits (e.g., image, status, recognition) could be related to pro-circular behaviors in different consumption stages and how these benefits could be used to develop circular business models.

We also stated that functional perceived value would positively influence repairing and reusing behaviors. Our findings supported these hypotheses for repairing practices in both categories (i.e., clothing and furniture) and reusing furniture. The reasoning here is that the functionality of products is needed to encourage repairing to extend the lifetime of products in their original status. For example, other studies on sustainable post-consumption clothing behaviors suggest that people are motivated to mend clothes when they intend to reuse them or give them to family, relatives, and friends [41]. This kind of reusing requires having clothing with proper features.

Regarding furniture, several purchases are driven by basic functional and utilitarian needs (e.g., comfort, design, quality, fair/reasonable price) [13,101,102]. Thus, when people perceive good functionality in their products, they could be encouraged to keep it by repairing them. Conversely, we found no significant relationship between functional value and reusing behavior in the clothing category. This is somewhat contradictory with previous studies that suggest people repair or reuse clothing by giving it to others [41]. A possible explanation is that there are several ways to practice reusing. For instance, people may reuse clothing through other behaviors like shopping for secondhand clothing [95] instead of reusing their own clothing. Because people find shopping fun and relaxing even in secondhand markets [95], they will not reuse their clothing even though such clothes have functional value.

Future research could particularly address reusing behavior, exploring different ways to reuse (e.g., donation, secondhand market, repurposing own products) and the role of functional value in all of them.

We found relevant evidence on the role of green consumption values in the relationship between perceived value and pro-circular behaviors. Notably, for the clothing category, the positive relationship between functional value and repairing was intensified for individuals with green consumption values, mainly for those consumers with a high perceived value. In addition, the relationship between social value and repairing became significant in the presence of green consumption values. Although we hypothesized that social value would have a negative relationship with repairing behavior, it seems that when people are concerned about the environment and reflect this concern with their consumption choices, the perceived social value of their products is transferred to procircular behavior, increasing the probability of repairing their clothes. These findings agree with previous studies that suggested that functional, social, and environmental values positively influence consumers' sustainable purchase behaviors [103]. Although there are few studies on pro-circular behaviors, some of these studies about mending also found a positive relationship between individuals' environmental concerns and the frequency of repairing [41,71]. Although previous research did not find support for the role of ethical concern as a mediating factor in the relationship between social value and green purchase intention [21], our findings suggest a possible new role for environmental concern in terms of green consumption values as a moderator or condition in the relationship between social value and sustainable behaviors; in this case, such behaviors are in favor of circularity like repairing. Nevertheless, the interaction among perceived social value, environmental values, and pro-circular behaviors deserves more attention. More research could be focused on these interrelationships.

Green consumption values were also important to intensify the effect of functional value on reusing behavior for clothing and furniture products. Consumers who have a high perception of the functionality of their products and additionally exhibit consumption values towards sustainability have a greater probability of reusing them always. Even though no previous studies explored the specific relationship that we analyzed, the relevance of green consumption values identified in our study is in line with previous research that highlights how green consumption values significantly and positively influence other pro-circular behaviors like reselling and reusing via donations [42,98]. Moreover, the findings suggest that consumers with green consumption values and high perceived functional value increase not only reusing behavior but also clothing repair behavior. In addition, the likelihood of clothing repair increases significantly for those consumers with high perceived social value because they translate their environmental concerns into their consumption decisions. Hence, our research adds to the previous literature by analyzing the role of green consumption values as a moderator between perceived value and pro-circular behaviors. We encourage future research to deepen this role to fully understand several relationships that help promote different circular business models.

Finally, we studied repairing and reusing as two behaviors with a similar purpose, like extending product lifespan. We analyzed these behaviors as separate practices. However, some studies mentioned that repairing could help individuals engage in other sustainable clothing consumption behaviors [104]. Some consumers repair some products to reuse them [41]. Previous research highlighted the spillovers between pro-environmental behaviors, some oriented to circularity [105]. Thus, future research could deeply study behavioral spillovers between pro-circular behaviors and analyze the role of perceived value and green consumption values in such relationships.

6. Conclusions

6.1. Managerial Implications

With circular economy principles in the strategic agendas of several brands, promoting circular business models, including repair and reuse services, may no longer be an option

but a requirement as a part of a brand's extended product responsibility (EPR) [106]. Additionally, it could be an opportunity for companies and consumers to act together, facing environmental issues by building a long-term relationship beyond the purchase stage. To foster these circular business models, companies can undertake several strategies from marketing, like promoting circular products and services, educating consumers, encouraging their active participation in circular behaviors, communicating circular values, and building a brand narrative that aligns with sustainable and circular practices. These strategies require understanding consumers, their current behaviors, values, and their relationship with the products they buy and use. This research aimed to examine all these factors by addressing a comprehensive analysis that explores the role of perceived value in adopting reusing and repairing behaviors and the interaction between products' valoration and environmental considerations in pro-circular behavior. The findings suggest valuable insights, allowing us to propose some managerial implications to promote circular business models, as we detail below:

Functional value is an essential starting point to encourage behaviors aiming to extend the lifetime of products. Our results suggest that when people perceive good functionality, durability, and quality in the products they buy, the probability of them repairing and reusing increases. In addition, current initiatives on repairing and reusing (e.g., Worn Wear from Patagonia; Decathlon Second Life) are supported in products designed and made to last. Hence, future strategies to promote circular business models based on extending the life cycle of products should be leveraged on joint work between marketing, design, and innovation teams to ensure proper product materials and features that ensure the properties needed to foster successful repairing and reusing models.

Enhancing emotional value once the product is purchased could be a great strategy to foster pro-circular behaviors of reusing and repairing. People could exhibit emotional attachment to products for several reasons (e.g., due to a special meaning, the service they provide, or the information they contain, in addition to some products being used for a long time) [7,107]. Because of this emotional connection, people may repair their damaged products [7] and mend to save loved garments [41]. Thus, brands oriented to circular business models could divert attention from perceived emotional value related to shopping towards emotional value focused on the phase of use, driving consumers to extend the lifespan of products.

Highlighting green consumption values could empower consumers and lead them towards circular business models. Our research showed that the interaction between sources of value increases the probability of reusing and repairing. Thus, whether brands focus their strategies on making the functional value of their products evident and reinforce the importance of consumer choices in tackling environmental issues, the acceptance of new business models such as reusing and repairing services could be higher.

There is a call to make sustainability a transversal area that works with all the functional areas in organizations. Marketing is one of the areas with a relevant role in sustainability issues, for example, through the promotion of new business models. The current study has a valuable impact by reinforcing this idea of a more systemic and comprehensive view by analyzing behaviors in stages beyond the purchase. Hence, our results and insights open new opportunities for marketing and sustainability working together. It implies the involvement of marketing from a strategic approach in which all marketing decisions (i.e., segmentation, targeting, positioning, and the whole marketing mix) have a more holistic perspective considering all the stages of consumption (i.e., acquisition, use, final disposal) and promoting pro-circular behaviors, hopefully using, and seizing the interaction between different sources of value, as our research suggests. By doing so, circular business models could help businesses differentiate themselves in the market and contribute to the broader goal of fostering a circular economy as a mandatory path to tackle social and environmental demands.

6.2. Limitations

This research was focused on two critical practices that aim to extend the useful life of products (i.e., reusing and repairing). Although the previous literature suggests these pro-circular behaviors are desirable options because of their potential environmental benefits across the whole value chain, other circular practices (e.g., resale, rent, recycling) deserve attention to fully understand which factors could be involved in promoting circular business models. Likewise, we used two product categories to operationalize pro-circular behaviors (i.e., furniture and clothing). Our purpose was not to analyze or compare the role of products in the studied behaviors. However, other product categories could be involved in future studies. Circularity is desirable in several industries, so more research, including more products, is needed.

This study based the analysis on seven neighborhoods in the city of Medellín, Colombia. Although this city is a reference point to study pro-circular behaviors, thanks to its initiatives and management in terms of environmental sustainability, the findings can only be inferred for this population. Further studies should be developed to expand the scope of these results, at least at the national level and hopefully globally.

Author Contributions: Conceptualization, C.A.; methodology, C.A., J.B.C.L. and M.A.B.B.; software, J.B.C.L. and M.A.B.B.; formal analysis, C.A., J.B.C.L. and M.A.B.B.; investigation, C.A.; data curation, J.B.C.L. and M.A.B.B.; writing—original draft preparation, C.A., J.B.C.L. and M.A.B.B.; writing—review and editing, C.A.; supervision, C.A.; project administration, C.A.; funding acquisition, C.A. All authors have read and agreed to the published version of the manuscript.

Funding: CESA School of Business funded the fieldwork of this research.

Institutional Review Board Statement: The study was approved by the Ethics Committee of CESA. (protocol code 004; 21 September 2021).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The raw data supporting the conclusions of this article will be made available by the authors on request.

Conflicts of Interest: The authors declare no conflicts of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

Appendix A

Table A1. Operationalization of variables.

Variable	Authors	Items	Scale	Cronbach α
Dependent Variables: Pro-circular Behaviors				
Repair	Scale adapted from Diddi and Yan [41]	 How often do you: Repair your furniture (e.g., chairs, tables, desks) instead of buying new ones. Repair your clothes instead of buying new ones. 	 Never Rarely Occasionally Often Always 	
Reuse	Scale adapted from Diddi and Yan [41]	How often do you: - Reuse your furniture (e.g., chairs, tables, desks). - Reuse your clothing.	 Never Rarely Occasionally Often Always 	

Table A1. Cont.

Variable	Authors	Items	Scale	Cronbach α
Independent Variables:				
Perceived value		Indicate your level of agreement with the following statements:		
Functional Value	Scale adapted from Saura and Vivó [79]	 Furniture The furniture I have (sofas, beds, chairs, desks, among others) is of an acceptable quality. The furniture I have (sofas, beds, chairs, desks, among others) is of an average level (functional but not sophisticated). The furniture I have (sofas, beds, chairs, desks, among others) will not last long. I feel that my furniture (sofas, beds, chairs, desks, among others) is good value for money. The furniture (sofas, beds, chairs, desks, among others) that I have will fulfill its functions properly. Clothing The clothes I have are of an acceptable quality. The clothes I have will not last long. I feel that my clothes are good value for money. 	 I strongly disagree I disagree I agree I strongly agree 	0.65
Emotional Value	Scale adapted from Saura and Vivó [79]	 Furniture I would feel good if I had new furniture (sofas, beds, chairs, desks, among others) at home. I would enjoy buying new furniture (sofas, beds, chairs, desks, among others). Buying new furniture (sofas, beds, chairs, desks, among others) would be to my liking. Clothing I would feel good if I had new clothes at home. I would enjoy buying new clothes. Buying new clothes would be to my liking. 		0.73
Social Value	Scale adapted from Saura and Vivó [79]	 Furniture Having new furniture (sofas, beds, chairs, desks, among others) would influence the image that others have of me. Buying new furniture (sofas, beds, chairs, desks, among others) would give a good impression of me to others. Clothing Having new clothes would influence the 		0.89
		image that others have of me.Buying new clothes would give a good impression of me to others.		0.94

Table A1. Cont.

Variable	Authors	Items	Scale	Cronbach α	
Moderator Variable: Green Consumption Values	Haws et al. [46]	 Indicate your level of agreement with the following statements: It is important to me that the products I use do not harm the environment. I consider the potential environmental impact of my actions when making many of my decisions. My buying habits are affected by my concern for our environment. I am concerned about the waste of resources on our planet. I would describe myself as environmentally responsible. I am willing to make myself uncomfortable to take actions that are more environmentally friendly. 	 I strongly disagree I disagree I agree I agree I strongly agree 	0.87	
Control Variables:					
Moral norms	Scale adapted from Vining and Ebreo [81]	 Indicate your level of agreement with the following statements: Furniture I feel a strong personal obligation to repair my furniture instead of buying new ones. I would feel guilty if instead of repairing my furniture I bought new ones. I am willing to go the extra mile to repair my furniture instead of buying new ones. I feel a strong personal obligation to repurpose my furniture (e.g., tables, chairs, desks) before discarding them. I would feel guilty if I did not repurpose my furniture (e.g., tables, chairs, desks) before discarding them. I am willing to go the extra mile to repurpose my furniture (e.g., tables, chairs, desks) before discarding them. 	 I strongly disagree I disagree I agree I strongly agree 	0.64 0.68	
		 Clothing I feel a strong personal obligation to repurpose my clothes instead of discarding them. I would feel guilty if I did not repurpose my clothes before discarding them. I am willing to go the extra mile to repurpose my clothes instead of discarding them. I feel a strong personal obligation to repair my clothes instead of buying new ones. I would feel guilty if instead of repairing my clothes I bought new ones. 		0.72	

Variable	Authors	Items	Scale	Cronbach α
Specific perceived effectiveness	Scale adapted from Vining and Ebreo [81]	 Indicate your level of agreement with the following statements: Furniture By reusing my furniture (e.g., tables, chairs, desks), I can contribute to solving environmental problems. By repairing my furniture (e.g., tables, chairs, desks), I can contribute to solving environmental problems. Clothing By reusing my clothing, I can contribute to solving environmental problems. By repairing my clothing, I can contribute to solving environmental problems. 	 I strongly disagree I disagree I agree I agree I strongly agree 	
Sociodemographics		Gender Age Marital status SES (Socio Economic Status) Occupation Level of studies Level of studies Level of income Type of housing		

Table A1. Cont.

Source: Own elaboration.

References

- 1. Koszewska, M. Circular Economy in Textiles and Fashion-the Role of a Consumer. In *Processing, Manufacturing, and Design;* Woodhead Publishing: Sawston, UK, 2018; pp. 183–206. [CrossRef]
- Bocken, N.M.P.; de Pauw, I.; Bakker, C.; van der Grinten, B. Product Design and Business Model Strategies for a Circular Economy. J. Ind. Prod. Eng. 2016, 33, 308–320. [CrossRef]
- Frishammar, J.; Parida, V. Circular Business Model Transformation: A Roadmap for Incumbent Firms. *Calif. Manag. Rev.* 2019, 61, 5–29. [CrossRef]
- 4. Kirchherr, J.; Reike, D.; Hekkert, M. Conceptualizing the Circular Economy: An Analysis of 114 Definitions. *Resour. Conserv. Recycl.* 2017, 127, 221–232. [CrossRef]
- Arias, C.; Beltrán, J.M.Q.; Ariza, J.M.M.; Lozano, J.B.C.; Bernal, M.A.B. Pro-Circular Consumer Profile: An Approach to Their Identification and Characterization Based on the Components of the Value-Belief-Norm Theory. *Sustainability* 2022, 14, 7883. [CrossRef]
- 6. Parajuly, K.; Fitzpatrick, C.; Muldoon, O.; Kuehr, R. Behavioral Change for the Circular Economy: A Review with Focus on Electronic Waste Management in the EU. *Resour. Conserv. Recycl. X* 2020, *6*, 100035. [CrossRef]
- Terzioğlu, N. Repair Motivation and Barriers Model: Investigating User Perspectives Related to Product Repair towards a Circular Economy. J. Clean. Prod. 2021, 289, 3–16. [CrossRef]
- Zeithaml, V.A. Consumer Perceptions of Price, Quality, and Value: A Means-End Model and Synthesis of Evidence. J. Mark. 1988, 52, 2–22. [CrossRef]
- 9. Peter, J.P.; Olson, J.C. Consumer Behavior and Marketing Strategy, 8th ed.; McGraw-Hill/Irwin: Boston, MA, USA, 2008; ISBN 978-0-07-352985-1.
- 10. Kim, I.; Jung, H.J.; Lee, Y. Consumers' Value and Risk Perceptions of Circular Fashion: Comparison between Secondhand, Upcycled, and Recycled Clothing. *Sustainability* **2021**, *13*, 1208. [CrossRef]
- 11. Campbell-Johnston, K.; Vermeulen, W.J.V.; Reike, D.; Brullot, S. The Circular Economy and Cascading: Towards a Framework. *Resour. Conserv. Recycl. X* 2020, *7*, 100038. [CrossRef]
- Sarath, P.; Bonda, S.; Mohanty, S.; Nayak, S.K. Mobile Phone Waste Management and Recycling: Views and Trends. *Waste Manag.* 2015, 46, 536–545. [CrossRef]
- Burnsed, K.A.; Hodges, N.J. Home Furnishings Consumption Choices: A Qualitative Analysis. *Qual. Mark. Res.* 2014, 17, 24–42. [CrossRef]
- Sánchez-Fernández, R.; Iniesta-Bonillo, M.Á. The Concept of Perceived Value: A Systematic Review of the Research. *Mark. Theory* 2007, 7, 427–451. [CrossRef]
- 15. Ashton, A.S.; Scott, N.; Solnet, D.; Breakey, N. Hotel Restaurant Dining: The Relationship between Perceived Value and Intention to Purchase. *Tour. Hosp. Res.* 2010, *10*, 206–218. [CrossRef]
- 16. Payne, A.; Holt, S. Diagnosing Customer Value: Integrating the Value Process and Relationship Marketing. *Br. J. Manag.* 2001, *12*, 159–182. [CrossRef]
- 17. Aksoy, R.; Basaran, U. The Effect of Perceived Value on Behavioural Intentions. Pressacademia 2017, 4, 1–16. [CrossRef]

- 18. Singh, S.; Singh, N.; Kalinić, Z.; Liébana-Cabanillas, F.J. Assessing Determinants Influencing Continued Use of Live Streaming Services: An Extended Perceived Value Theory of Streaming Addiction. *Expert Syst. Appl.* **2021**, *168*, 114241. [CrossRef]
- 19. Sinha, S.K.; Verma, P. Impact of Sales Promotion's Benefits on Perceived Value: Does Product Category Moderate the Results? J. Retail. Consum. Serv. 2020, 52, 101887. [CrossRef]
- Lee, S.; Sung, B.; Phau, I.; Lim, A. Communicating Authenticity in Packaging of Korean Cosmetics. J. Retail. Consum. Serv. 2019, 48, 202–214. [CrossRef]
- 21. Suphasomboon, T.; Vassanadumrongdee, S. Toward Sustainable Consumption of Green Cosmetics and Personal Care Products: The Role of Perceived Value and Ethical Concern. *Sustain. Prod. Consum.* **2022**, *33*, 230–243. [CrossRef]
- 22. Sweeney, J.C.; Soutar, G. Consumer Perceived Value: The Development of a Multiple Item Scale. J. Retail. 2001, 7, 203–220. [CrossRef]
- 23. Sheth, J.N.; Newman, B.I.; Gross, B.L. Why We Buy What We Buy: A Theory of Consumption Values: Discovery Service for Air Force Institute of Technology. *J. Bus. Res.* **1991**, 22, 159–170. [CrossRef]
- Higueras-Castillo, E.; Molinillo, S.; Coca-Stefaniak, J.A.; Liébana-Cabanillas, F. Perceived Value and Customer Adoption of Electric and Hybrid Vehicles. Sustainability 2019, 11, 4956. [CrossRef]
- Yu, S.; Lee, J. The Effects of Consumers' Perceived Values on Intention to Purchase Upcycled Products. Sustainability 2019, 11, 1034. [CrossRef]
- 26. Zhao, S.; Chen, L. Exploring Residents' Purchase Intention of Green Housings in China: An Extended Perspective of Perceived Value. *Int. J. Environ. Res. Public. Health* **2021**, *18*, 4074. [CrossRef] [PubMed]
- 27. De Medeiros, J.F.; Ribeiro, J.L.D.; Cortimiglia, M.N. Influence of Perceived Value on Purchasing Decisions of Green Products in Brazil. *J. Clean. Prod.* 2016, *110*, 158–169. [CrossRef]
- 28. Danish, M.; Ali, S.; Ahmad, M.A.; Zahid, H. The Influencing Factors on Choice Behavior Regarding Green Electronic Products: Based on the Green Perceived Value Model. *Economies* **2019**, *7*, 99. [CrossRef]
- 29. Hsu, S.Y.; Chang, C.C.; Lin, T.T. Triple Bottom Line Model and Food Safety in Organic Food and Conventional Food in Affecting Perceived Value and Purchase Intentions. *Br. Food J.* **2019**, *121*, 333–346. [CrossRef]
- Lim, W.M.; Yong, J.L.S.; Suryadi, K. Consumers' Perceived Value and Willingness to Purchase Organic Food. J. Glob. Mark. 2014, 27, 298–307. [CrossRef]
- Zhang, Y.; Xiao, C.; Zhou, G. Willingness to Pay a Price Premium for Energy-Saving Appliances: Role of Perceived Value and Energy Efficiency Labeling. J. Clean. Prod. 2020, 242, 118555. [CrossRef]
- 32. Wang, Y.; Liu, X.; Huang, M.; Zuo, J.; Rameezdeen, R. Received vs. given: Willingness to Pay for Sponge City Program from a Perceived Value Perspective. *J. Clean. Prod.* 2020, 256, 120479. [CrossRef]
- Park, E.; Kwon, S.J. What Motivations Drive Sustainable Energy-Saving Behavior?: An Examination in South Korea. *Renew. Sustain. Energy Rev.* 2017, 79, 494–502. [CrossRef]
- Confente, I.; Scarpi, D.; Russo, I. Marketing a New Generation of Bio-Plastics Products for a Circular Economy: The Role of Green Self-Identity, Self-Congruity, and Perceived Value. J. Bus. Res. 2020, 112, 431–439. [CrossRef]
- 35. Llerena, D. Green Consumer Behaviour: An Experimental. Bus. Strategy Environ. 2011, 420, 408–420.
- 36. Harms, R.; Linton, J.D. Willingness to Pay for Eco-Certified Refurbished Products: The Effects of Environmental Attitudes and Knowledge. *J. Ind. Ecol.* **2016**, *20*, 893–904. [CrossRef]
- 37. Wallner, T.S.; Magnier, L.; Mugge, R. An Exploration of the Value of Timeless Design Styles for the Consumer Acceptance of Refurbished Products. *Sustainability* **2020**, *12*, 1213. [CrossRef]
- Bundgaard, A.M.; Huulgaard, R.D. Luxury Products for the Circular Economy? A Case Study of Bang & Olufsen. Bus. Strategy Environ. 2019, 28, 699–709. [CrossRef]
- 39. Testa, F.; Gusmerotti, N.; Corsini, F.; Bartoletti, E. The Role of Consumer Trade-Offs in Limiting the Transition towards Circular Economy: The Case of Brand and Plastic Concern. *Resour. Conserv. Recycl.* **2022**, *181*, 106262. [CrossRef]
- 40. Mostaghel, R.; Chirumalla, K. Role of Customers in Circular Business Models. J. Bus. Res. 2021, 127, 35–44. [CrossRef]
- Diddi, S.; Yan, R.N. Consumer Perceptions Related to Clothing Repair and Community Mending Events: A Circular Economy Perspective. Sustainability 2019, 11, 5306. [CrossRef]
- Tan, T.M.; Makkonen, H.; Kaur, P.; Salo, J. How Do Ethical Consumers Utilize Sharing Economy Platforms as Part of Their Sustainable Resale Behavior? The Role of Consumers' Green Consumption Values. *Technol. Forecast. Soc. Chang.* 2022, 176, 121432. [CrossRef]
- 43. Camacho-Otero, J.; Tunn, V.; Chamberlin, L.; Boks, C. Consumers in the Circular Economy. In *Handbook of the Circular Economy*; Edward Elgar Publishing: Cheltenham, UK, 2020; pp. 74–87.
- 44. Mugge, R. Product Design and Consumer Behaviour in a Circular Economy. Sustainability 2018, 10, 3704. [CrossRef]
- 45. Wastling, T.; Charnley, F.; Moreno, M. Design for Circular Behaviour: Considering Users in a Circular Economy. *Sustainability* **2018**, *10*, 1743. [CrossRef]
- 46. Haws, K.L.; Winterich, K.P.; Naylor, R.W. Seeing the World through GREEN-Tinted Glasses: Green Consumption Values and Responses to Environmentally Friendly Products. *J. Consum. Psychol.* **2014**, *24*, 336–354. [CrossRef]

- 47. Carrington, M.J.; Neville, B.A.; Whitwell, G.J. Why Ethical Consumers Don't Walk Their Talk: Towards a Framework for Understanding the Gap between the Ethical Purchase Intentions and Actual Buying Behaviour of Ethically Minded Consumers. *J. Bus. Ethics* **2010**, *97*, 139–158. [CrossRef]
- 48. Shaw, D.; McMaster, R.; Newholm, T. Care and Commitment in Ethical Consumption: An Exploration of the 'Attitude–Behaviour Gap'. J. Bus. Ethics **2016**, 136, 251–265. [CrossRef]
- 49. Bailey, A.A.; Mishra, A.S.; Tiamiyu, M.F. Application of GREEN Scale to Understanding US Consumer Response to Green Marketing Communications. *Psychol. Mark.* 2018, *35*, 863–875. [CrossRef]
- 50. Gan, C.; Wang, W. The Influence of Perceived Value on Purchase Intention in Social Commerce Context. *Internet Res.* 2017, 27, 772–785. [CrossRef]
- 51. Rintamäki, T.; Kanto, A.; Kuusela, H.; Spence, M.T. Decomposing the Value of Department Store Shopping into Utilitarian, Hedonic and Social Dimensions: Evidence from Finland. *Int. J. Retail Distrib. Manag.* **2006**, *34*, 6–24. [CrossRef]
- 52. Chiu, C.M.; Wang, E.T.G.; Fang, Y.H.; Huang, H.Y. Understanding Customers' Repeat Purchase Intentions in B2C e-Commerce: The Roles of Utilitarian Value, Hedonic Value and Perceived Risk. *Inf. Syst. J.* **2014**, *24*, 85–114. [CrossRef]
- 53. Hsu, C.L.; Lin, J.C.C. What Drives Purchase Intention for Paid Mobile Apps?—An Expectation Confirmation Model with Perceived Value. *Electron. Commer. Res. Appl.* 2015, 14, 46–57. [CrossRef]
- 54. Hu, T.; Kettinger, W.J.; Poston, R.S. The Effect of Online Social Value on Satisfaction and Continued Use of Social Media. *Eur. J. Inf. Syst.* 2015, 24, 391–410. [CrossRef]
- Kim, S.B.; Sun, K.A.; Kim, D.Y. The Influence of Consumer Value-Based Factors on Attitude-Behavioral Intention in Social Commerce: The Differences between High- and Low-Technology Experience Groups. J. Travel Tour. Mark. 2013, 30, 108–125. [CrossRef]
- 56. Li, M.; Dong, Z.Y.; Chen, X. Factors Influencing Consumption Experience of Mobile Commerce: A Study from Experiential View. *Internet Res.* **2012**, *22*, 120–141. [CrossRef]
- 57. Lin, K.-Y.; Lu, H.-P. Predicting Mobile Social Network Acceptance Based on Mobile Value and Social Influence. *Internet Res.* 2015, 25, 107–130. [CrossRef]
- Amatulli, C.; De Angelis, M.; Donato, C. An Investigation on the Effectiveness of Hedonic versus Utilitarian Message Appeals in Luxury Product Communication. *Psychol. Mark.* 2020, 37, 523–534. [CrossRef]
- 59. Wiedmann, K.; Hennigs, N.; Siebels, A. Value-based Segmentation of Luxury Consumption Behavior. *Psychol. Mark.* 2009, 26, 625–651. [CrossRef]
- Ahmad, S.N.B.; Nadzri, N.M.; Shaari, A.; Yunus, S.; Musa, N.B.C. Perceived Values and Personal Values: Which One Explains the Consumer's Repurchase Intention of Eco-Friendly Home Appliances Product? *Int. J. Innov. Creat. Chang.* 2019, *6*, 268–284. [CrossRef]
- 61. Lou, X.; Chi, T.; Janke, J.; Desch, G. How Do Perceived Value and Risk Affect Purchase Intention toward Second-Hand Luxury Goods? An Empirical Study of U.S. Consumers. *Sustainability* **2022**, *14*, 11730. [CrossRef]
- 62. Butcher, K.; Sparks, B.; O'Callaghan, F. Effect of Social Influence on Repurchase Intentions. J. Serv. Mark. 2002, 16, 503–514. [CrossRef]
- Roig, J.C.F.; García, J.S.; Tena, M.Á.M. Perceived Value and Customer Loyalty in Financial Services. Serv. Ind. J. 2009, 29, 775–789.
 [CrossRef]
- 64. Bhattacherjee, A. Understandinignformatiosnystems Continuancea: An Expectation-Confirmation Model. *MIS Q.* 2001, 25, 351–370. [CrossRef]
- 65. Hsu, M.H.; Chiu, C.M.; Ju, T.L. Determinants of Continued Use of the WWW: An Integration of Two Theoretical Models. *Ind. Manag. Data Syst.* **2004**, *104*, 766–775. [CrossRef]
- Westbrook, R.A.; Oliver, R.L. The Dimensionality of Consumption Emotion Patterns and Consumer Satisfaction. J. Consum. Res. 1991, 18, 84. [CrossRef]
- 67. Alan, A.K.; Dursun, I.; Kabadayi, E.T.; Aydin, K.; Anlagan, F. What Influences the Repurchase Intention for Luxury Brands?—The Relative Impacts of Luxury Value Dimensions. *Int. Bus. Res.* **2016**, *9*, 11. [CrossRef]
- Young, J.H.; Nunes, J.C.; Drèze, X. Signaling Status with Luxury Goods: The Role of Brand Prominence. J. Mark. 2010, 74, 15–30.
 [CrossRef]
- 69. Kunamaneni, S.; Jassi, S.; Hoang, D. Promoting Reuse Behaviour: Challenges and Strategies for Repeat Purchase, Low-Involvement Products. *Sustain. Prod. Consum.* **2019**, *20*, 253–272. [CrossRef]
- 70. Evans, S.; Cooper, T. Consumer Influences on Product Life-Spans In Longer Lasting Products, 1st ed.; Routledge: New York, NY, USA, 2010; ISBN 978-1-315-59293-0.
- 71. Laitala, K.; Klepp, I.G. Care and Production of Clothing in Norwegian Homes: Environmental Implications of Mending and Making Practices. *Sustainability* **2018**, *10*, 2899. [CrossRef]
- Barr, S.; Gilg, A.W.; Ford, N.J. Differences between Household Waste Reduction, Reuse and Recycling Behaviour: A Study of Reported Behaviours, Intentions and Explanatory Variables. *Environ. Waste Manag.* 2001, 4, 69–82.
- 73. Gullstrand Edbring, E.; Lehner, M.; Mont, O. Exploring Consumer Attitudes to Alternative Models of Consumption: Motivations and Barriers. J. Clean. Prod. 2016, 123, 5–15. [CrossRef]

- 74. Morgan, L.R.; Birtwistle, G. An Investigation of Young Fashion Consumers' Disposal Habits. *Int. J. Consum. Stud.* 2009, 33, 190–198. [CrossRef]
- 75. Velandia, E. Medellín es un Referente Internacional en Movilidad Eléctrica 2018. Available online: https://www.metropol.gov.co/ Paginas/Noticias/elmetropolitano-entrevistas/medellin-es-un-referente-internacional-en-movilidad-electrica-edder-v.aspx (accessed on 15 October 2023).
- 76. Sáenz, J.D. Por Invitación de Naciones Unidas, Medellín Se Suma al Resto Del Mundo Como Una Ciudad Verde; Alcaldia Medellín: Medellín, Colombia, 2020.
- Alcaldía de Medellín. Indicador Multidimensional de Condiciones de Vida 2013; Alcaldía de Medellín: Medellín, Colombia, 2013; pp. 1–6.
- 78. Murray, R.S.; Larry, J.S. Estadistica, 4th ed.; McGraw Hill México: Mexico City, Mexico, 2015; ISBN 978-1-78728-439-5.
- 79. Sales Vivó, V.; Gil Saura, I. Valor Percibido Por El Consumidor: Una Aplicación En La Compra de Equipamiento Para El Hogar. *Estud. Sobre Consumo* **2007**, *82*, 35–48.
- 80. Stern, P.C.; Dietz, T.; Abel, T.; Guagnano, G.A.; Kalof, L. A Value-Belief-Norm Theory of Support for Social Movements: The Case of Environmentalism. *Hum. Ecol. Rev.* **1999**, *6*, 81–97.
- 81. Vining, J.; Ebreo, A. Predicting Recycling Behavior from Global and Specific Environmental Attitudes and Changes in Recycling Opportunities. *J. Appl. Soc. Psychol.* **1992**, 22, 1580–1607. [CrossRef]
- 82. Ellen, P.; Wiener, J.L.J.; Cobb-Walgren, C.; Scholder Ellen, P.; Wiener, J.L.J.; Cobb-Walgren, C. The Role of Perceived Consumer Effectiveness in Motivating Environmentally Conscious Behaviors. *J. Public Policy Mark.* **1991**, *10*, 102–117. [CrossRef]
- Scott, D. Equal Opportunity, Unequal Results: Determinants of Household Recycling Intensity. *Environ. Behav.* 1999, 31, 267–290.
 [CrossRef]
- 84. Antil, J.H. Socially Responsible Consumers: Profile and Implications for Public Policy. J. Macromark. 1984, 4, 18–39. [CrossRef]
- Ursachi, G.; Horodnic, I.A.; Zait, A. How Reliable Are Measurement Scales? External Factors with Indirect Influence on Reliability Estimators. *Procedia Econ. Financ.* 2015, 20, 679–686. [CrossRef]
- Raharjanti, N.W.; Wiguna, T.; Purwadianto, A.; Soemantri, D.; Indriatmi, W.; Poerwandari, E.K.; Mahajudin, M.S.; Nugrahadi, N.R.; Roekman, A.E.; Saroso, O.J.D.A.; et al. Translation, Validity and Reliability of Decision Style Scale in Forensic Psychiatric Setting in Indonesia. *Heliyon* 2022, 8, e09810. [CrossRef]
- 87. StataCorp. Stata 18 Base Reference Manual; Stata Press: College Station, TX, USA, 2023; Volume 18, ISBN 978-1-59718-098-6.
- 88. Nordberg, L. Stepwise Selection of Explanatory Variables in the Binary Logit Model. Scand. J. Stat. 1981, 8, 17–26.
- 89. Soroush, A.; Bahreininejad, A.; Van Den Berg, J. A Hybrid Customer Prediction System Based on Multiple Forward Stepwise Logistic Regression Mode. *Intell. Data Anal.* **2012**, *16*, 265–278. [CrossRef]
- 90. Saunders, D.R. Moderator Variables in Prediction. Educ. Psychol. Meas. 1956, 16, 209–222. [CrossRef]
- Sharma, S.; Durand, R.M.; Gur-Arie, O. Identification and Analysis of Moderator Variables. *J. Mark. Res.* 1981, *18*, 291. [CrossRef]
 Dawson, J.F. Moderation in Management Research: What, Why, When, and How. *J. Bus. Psychol.* 2014, *29*, 1–19. [CrossRef]
- Dawson, J.F. Moderation in Management Research: What, Why, When, and How. J. Bus. Psychol. 2014, 29, 1–19. [CrossRef]
 Mclaren, A.; Mclauchlan, S. Crafting Sustainable Repairs: Practiced-based Approaches To Extending the Life of Clothes. In Proceedings of the Product Lifetimes And The Environment-PLATE Conference, Nottingham Trent University, Nottingham, UK, 17–19 June 2015.
- 94. Niinimäki, K.; Hassi, L. Emerging Design Strategies in Sustainable Production and Consumption of Textiles and Clothing. J. Clean. Prod. 2011, 19, 1876–1883. [CrossRef]
- Silva, C.; Mccarver, M.; Wang, C.-H.; Wang, T.-Y.; Ting, C.U.S. Generation Z Consumers' Motivations and Purchase Behavior Towards Secondhand Clothing. In Proceedings of the International Textile and Apparel Association Annual Conference, Virtual, 18–20 November 2020.
- 96. Csikszentmihalyi, M.; Rochberg-Halton, E. The Meaning of Things: Domestic Symbols and the Self. In *Contemporary Sociology*; Cambridge University Press: Cambridge, UK, 1981; Volume 12.
- Ladik, D.; Carrillat, F.; Tadajewski, M. Belk's (1988) "Possessions and the Extended Self" Revisited. J. Hist. Res. Mark. 2015, 7, 184–207. [CrossRef]
- 98. Lai, C.C.; Chang, C.E. Clothing Disposal Behavior of Taiwanese Consumers with Respect to Environmental Protection and Sustainability. *Sustainability* **2020**, *12*, 9445. [CrossRef]
- 99. Pepper, M.; Jackson, T.; Uzzell, D. An Examination of the Values That Motivate Socially Conscious and Frugal Consumer Behaviours. *Int. J. Consum. Stud.* 2009, 33, 126–136. [CrossRef]
- 100. Trujillo, C.A. The Future of Sustainable Consumption after the Pandemic, Optimism or Pessimism? In A New Era of Consumer Behavior-Beyond the Pandemic; IntechOpen: London, UK, 2022.
- 101. Wang, Q.; Shi, G.; Chan-Haldbrendt, C. Market Potential for Fine Furniture Manufactured from Low-Grade Hardwood: Evidence from a Conjoint Analysis in the Northeastern United States. *For. Prod. J.* **2004**, *54*, 19–25.
- 102. Bennington, R.R. Furniture Marketing: From Product Development to Distribution, 2nd ed.; Fairchild Publications, Inc.: New York, NY, USA, 2004.
- Khan, S.N.; Mohsin, M. The Power of Emotional Value: Exploring the Effects of Values on Green Product Consumer Choice Behavior. J. Clean. Prod. 2017, 150, 65–74. [CrossRef]

- 104. Diddi, S.; Yan, R.N.; Bloodhart, B.; Bajtelsmit, V.; McShane, K. Exploring Young Adult Consumers' Sustainable Clothing Consumption Intention-Behavior Gap: A Behavioral Reasoning Theory Perspective. Sustain. Prod. Consum. 2019, 18, 200–209. [CrossRef]
- 105. Arias, C.; Trujillo, C.A. Perceived Consumer Effectiveness as a Trigger of Behavioral Spillover Effects: A Path towards Recycling. *Sustainability* **2020**, *12*, 4348. [CrossRef]
- 106. Kant Hvass, K.; Pedersen, E.R.G. Toward Circular Economy of Fashion: Experiences from a Brand's Product Take-Back Initiative. *J. Fash. Mark. Manag.* 2019, 23, 345–365. [CrossRef]
- 107. Chapman, J. Design for (Emotional) Durability. Des. Issues 2009, 25, 29-35. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.